SYMPTOMS OF ADJUSTMENT DISORDER AND SMOKING PREDICT LONG-TERM FUNCTIONAL OUTCOME AFTER ANKLE AND LOWER LEG FRACTURES

Sabine WEIMERT, MSc¹, Sebastian KUHN, MD, PhD², Pol M. ROMMENS, MD, PhD², Manfred E. BEUTEL, MD, PhD¹ and Iris C. REINER, PhD¹

From the ¹Department of Psychosomatic Medicine and Psychotherapy, University Medical Center of the Johannes Gutenberg University Mainz, and ²Department of Orthopaedics and Traumatology, University Medical Center of the Johannes Gutenberg University Mainz, Mainz, Germany

Objective: To examine the psychological, social, behavioural and injury-related factors impacting functional outcome in patients with ankle or lower limb fracture one year post-operation.

Methods: In this prospective study 66 patients with ankle or lower leg fracture were recruited and followed up one year post-operation. Possible associations between predictors and functional outcome were explored by regression analyses. Functional outcome was assessed with the American Orthopedic Foot and Ankle Society Ankle-Hindfoot Score.

Results: Linear regression models revealed that smoking and elevated symptoms of adjustment disorder were associated with lower functional outcome one year post-operation. Fracture type, depressive symptoms and quality of relationship had no effect on functional outcome. A second linear regression revealed that preoccupations were correlated with functional outcome.

Conclusion: Smoking and symptoms of adjustment disorder, specifically preoccupations, are associated with functional outcome one year post-operation in patients recovering from ankle or lower leg fractures. The results support the notion that differences in functional recovery are attributable to psychological and behavioural factors rather than to fracture type. Psychological, fracture-specific, symptoms play a role in functional recovery rather than general affective symptoms.

Key words: ankle fractures; functional outcome; recovery of function; rehabilitation; adjustment; psychosocial factors.

Accepted May 20, 2020; Epub ahead of print Jun 4, 2020

J Rehabil Med 2020; 52: jrm00086

Correspondence address: Sabine Weimert, Universitätsmedizin Mainz, Klinik und Poliklinik für Psychosomatische Medizin und Psychotherapie, Untere Zahlbacher Str. 8, 55131 Mainz, Germany. E-mail: sweimert@ students.uni-mainz.de

Lower leg and ankle fractures remain one of the most common injuries treated by orthopaedic surgeons (1, 2). Studies suggest that long-term impairments persist up to 3 years after ankle fractures, including pain, stiffness, impaired activity level and function (3-5).

Smoking and injury characteristics are biological factors associated with poor outcomes of operatively treated ankle or lower leg fractures. Smoking is asso-

LAY ABSTRACT

After ankle and lower leg fractures, some patients remain impaired for up to one year. Psychosocial factors are known to influence recovery after orthopaedic trauma. This long-term study examined how psychosocial, behavioural and injury-related factors are associated with functional outcome one year after injury. The results show that patients who smoke at 4 weeks after the fracture and have increased symptoms of adjustment disorder have a poorer outcome one year after injury. Fracture type, social support in the relationship and depressive symptoms are not associated with outcome one year after the fracture.

ciated with more complications (6) and lower physical functioning up to 3 months after surgical treatment (5). Weber C fractures are associated with poorer functional outcomes than less disruptive Weber A fractures (7).

Psychosocial factors may account for a large proportion of the variance in functioning in orthopaedic conditions (8, 9) and after orthopaedic trauma (10), e.g. relationship conditions (11). Perceived social support in relationships is crucial concerning health outcome after injury (12, 13). Patients with ankle fractures commonly report psychological distress (14). Among patients with lower limb fractures, 32-41% experience depression and notably, depressive symptoms are associated with poorer functional outcome (15, 16). Nota et al. found that depressive symptoms shortly after orthopaedic trauma were correlated with disability 5-8 months after trauma (9). When patients struggle to adapt to these changed circumstances it can result in psychological disorders, such as adjustment disorder (AD). Conceptually, AD is an intermediate health condition between normal responses to stress and more severe psychological disorders, such as anxiety and depression (17). AD overlaps with anxiety and depression, but comprises heterogeneous symptoms, such as rumination, disorders of social behaviour or irritability. To our knowledge there is no prospective study investigating the role of symptoms of AD in the context of fractures, despite the fact that AD is one of the most common used psychological diagnoses in psychiatry (18) and in the medical context (19–21). Comprehensive follow-up of symptoms of this frequent psychological disorder are necessary after ankle and lower leg fractures.

p. 2 of 7 S. Weimert et al.

The aims of this prospective observational study were: (*i*) to determine the impact of biological (fracture type, smoking), social (social support from partner) and psychological variables (symptoms of depression and AD) at 4 weeks post-operation on functional outcome one year post-operation, after accounting for baseline functional outcome and age; and, since it was assumed that symptoms of AD would co-vary with longterm functional outcome; and (*ii*) to explore which symptoms of AD are specifically related to functional outcome one year post-operation.

MATERIAL AND METHODS

Sample and recruitment

Under an institutional review board-approved protocol, patients who sustained a fracture of the ankle or the lower leg and who were treated at the University Hospital of a large public university in Germany between 1 December 2014 and 31 April 2016 were assessed for eligibility.

Inclusion criteria were: (*i*) age 18 years or older; (*ii*) fracture of the lower leg (AO 41-A, 41-C, 42-A, 42-B, 42-C, 43-A, 43-B, 43-C) or the ankle (AO 44-A, 44-B, 44-C) that required surgery; (*iii*) sufficient command of the German language; and (*iv*) Injury Severity Score (ISS) lower than 16 on admission.

Exclusion criteria were: (i) malignant disease; (ii) pathological fractures (e.g. caused through osteoporosis); (iii) psychosis, or active alcohol or substance dependence; (iv) cognitive deficiency or other factors that could interfere with informed consent processes; (v) pregnancy or breastfeeding. Some fractures, such as uncomplicated ankle fracture, might be treated with conservative treatments, but for better comparability, the study included only those patients who were treated operatively. Patients who sustained an AO type 41 B fracture were not included, since they were already included concurrently in another study at the same clinic. Eligible patients were approached shortly after their operation. Patients who agreed to participate completed a questionnaire measuring symptoms of AD, mechanism of injury and pre-existing psychological disorders at enrollment (Baseline). Since the study focuses on functional outcome, the patients for whom the data on functional outcome were available at both 4 weeks post-operation and at follow-up, formed the study cohort (Table I). There were no significant differences in age, sex, symptoms of AD or fracture type between patients with and without complete follow-up data. A flow diagram of the patients who completed both assessments of functional outcome is presented in Fig. 1. The patients were recruited, on average, 2 days after the operation.

At 4 weeks post-operation, patients were physically examined by an intensively trained and experienced medical student to assess functional outcome, and interviewed by a psychologist to assess psychological disorders. This was done as an outpatient assessment. If the patient had not been discharged by 4 weeks post-operation, the examination and interview was performed in the patient's hospital room. Physical examination included the measurement of flexion and extension angles and the circumference of the knee joint gap, ankle and lower leg. In addition, a standardized procedure for the measurement of functional outcome was performed. At 4 weeks post-operation, patients also completed a set of questionnaires that contained demographic questions, questions on smoking behaviour and validated measures of symptoms of AD, depressive symptoms and relationship quality. Six months post-operation patients completed a set of questionnaires on the symptoms of AD and depression. A follow-up assessment was conducted one year post-operation: patients underwent physical examination and completed questionnaires on symptoms of AD and depression. Mechanism of injury, fracture type and ISS were abstracted from the medical record.

The time frame of 4 weeks was based on our experience that recovery (e.g. taking little or no pain medication, bearing weight) is not yet established by this time. After that time we expected the onset of symptoms of AD, since patients left the hospital, where they experienced full-time care, and had to manage everyday life in their regular environment. In contrast, by one year postoperation recovery is well established in most patients. Since this study is observational, patients who were diagnosed with AD were provided with information about treatment options.

The study was approved by the ethics committee of the State Board of Physicians of Rhineland-Palatinate (Germany). All participants provided written informed consent to participate in this study.

 $\ensuremath{\textbf{Table I.}}$ Demographic variables and baseline characteristics of the study cohort

the study cohort	
Variable	<i>n</i> = 66
Age, years, mean (SD) [range]	41.9 (15.2) [18-73]
Male, % (n)	53 (35)
Educational level, % (n)	
Lower secondary education	19.7 (13)
Secondary education	22.7 (15)
High school	27.3 (18)
University	30.3 (20)
Marital status, % (n)	
Single	42.4 (28)
Married	48.5 (32)
Divorced/separated	9.1 (6)
Relationship status, % (n)	
In relationship	75.8 (50)
Working status, % (n)	
Employed	57.6 (38)
Homemaker	1.5 (1)
Retired	9.1 (6)
In education	15.2 (10)
Currently on sick leave	16.7 (11)
Circumstances of injury, % (n)	
Work	3.0 (2)
Traffic	9.1 (6)
Sports	19.7 (13)
Free time	68.2 (45)
Fracture type (AO Classification), % (n)	
42A	3.0 (2)
44A	3.0 (2)
42B	4.5 (3)
43B	1.5 (1)
44B ^a	60.6 (40)
41C	3.0 (2)
42C	4.5 (3)
43C	4.5 (3)
44C ^b	13.6 (9)
Length of hospital stay, days, mean (SD) [range]	8.7 (6.4) [3-41]
Injury Severity Score, mean (SD) [range]	4.2 (1.2) [4-13]
Smoking behaviour	() []
Smokers, % (n)	50.8 (34)
Cigarettes per day, mean (SD) [range]	8.8 (12.1) [0-60]
Quality of relationship, mean (SD) [range]	
QRI Social Support, mean (SD)	3.47 (0.41)

^aThe patient also sustained an AO type 42A1 fracture on the same leg. ^bOne patient sustained the same fracture (type) on both legs.



Fig. 1. Flowchart of patients who completed both assessments of functional outcome (American Orthopedic Foot and Ankle Society; AOFAS).

Measurements

Fracture type. To characterize the fracture type, X-ray or computed tomography (CT) images of the fractures were reviewed and classified according to the Mueller AO Classification by an orthopaedic surgeon (22). For further analysis fractures were categorized according to the sub-classification, resulting in 3 groups: AO type A, B and C.

Functional outcome. Functional outcome was measured with the American Orthopedic Foot and Ankle Society (AOFAS) Ankle-Hindfoot Score (23). The AOFAS Score is among the most commonly used instruments for measuring treatment outcome in patients who have sustained an ankle or hindfoot injury. It comprises a patient-reported part concerning pain and function and an objective part based on the clinicians' physical examination of the patient (e.g. assessing sagittal motion, ankle-hindfoot stability). The scale includes 9 items that can be divided into 3 subscales: pain (ranging from 0-40), function (ranging from 0-50) and alignment (ranging from 0-10). The sum score can range from 0 to 100. Higher scores indicate no impairments. A sum score of 90 or above is considered "excellent", scores of 80 to 89 "good", 60 to 79 "fair" and lower scores are considered as "poor" (23). The validity of the AOFAS Score and the reliability of the patient-reported part have been confirmed (24, 25). The reliability of the objective part has not yet been confirmed.

Symptoms of an adjustment disorder

Symptoms of an AD were measured with the Adjustment Disorder New Module (ADNM) (26). The self-administered questionnaire consists of 29 items, which ask about the frequency of symptoms that the patients experienced on a scale ranging from 0 to 3. The ADNM was adjusted so that all items referred to the accident that caused the fracture. The sum score can range from 29 to 116, with high scores indicating more symptoms of AD. Six subscales can be evaluated: intrusions, avoidance, failure to adapt, anxiety, depression and disorders of impulse control. The scales are not standardized. After the ADNM was conceptualized, the model of AD on which it is based on, as well as the ADNM itself, has been further developed. The symptom cluster, which was called "intrusions" has been developed to "preoccupations" (27). The current terminology is therefore adopted. Reliability and validity have been confirmed (28). The internal consistency of the sum score was high (Cronbach's α 0.95) in the current sample.

Depressive symptoms. Depressive symptoms were measured with the German Version of the Patient Health Questionnaire -9 (PHQ-9) (29). The self-administered questionnaire consists of 9 items regarding the frequency of depressive symptoms. Items are scored from 0 to 3, with a maximum score of 27. A score of 10 or above is frequently used as a cut-off point to identify patients with major depression. The PHQ-9 has been proven to be a reliable and valid measure of depressive symptoms (30). The internal consistency of this was satisfactory (Cronbach's α 0.76) in the current sample. One patient did not provide information on depressive symptoms.

Psychological disorders. To assess psychological disorders and history of mental illness the German Version of the Structured Clinical Interview for the Diagnostic and Statistical Manual of Mental Disorders was conducted (SCID-I and SCID-II) (31).

Social support. The Quality of Relationships Inventory (QRI) (32) is a self-administered questionnaire that consists of 25 items, scored from 1 to 4. The QRI is divided into 3 scales: Social Support, Conflict and Depth. The QRI has been shown to be reliable and valid (32). The internal consistency of the Social Support Scale was satisfactory (Cronbach's $\alpha 0.77$) in the current sample.

Fifty patients completed the QRI, 16 patients were not currently in a relationship. Only the social support scale was used for further analysis.

Smoking behaviour. Information on smoking behaviour and history was determined by the following questions: (*i*) "Do you smoke?" (*ii*) "If yes: I've been smoking since ... years" (*iii*) "How many cigarettes per day do you smoke?". One patient did not provide information about smoking behaviour. The number of cigarettes smoked per day was included in the regression analysis.

Statistical analysis

Descriptive statistics were used to describe both demographic and primary study variables. χ^2 tests, a Mann–Whitney U test and analyses of variance (ANOVAs) were used to compare drop-outs with those who completed the follow-up in terms of demographic data, fracture type and ADNM scores. Measures of functional outcome, quality of relationship, and symptoms of AD and depression were treated as continuous scales. Pearson's correlation coefficients were computed to indicate the strength and direction of relationship between the 2 variables. Fracture type was treated as an ordinal scale and Spearman's rho was calculated to indicate the strength and direction of relationships between fracture type and other variables. Changes in ADNM scores over time were evaluated with repeated measures analysis of variance (rmA-NOVA). As the data violated the assumption of sphericity, the Greenhouse-Geisser correction was applied. Change in AOFAS scores over time were evaluated with a *t*-test for paired samples. In order to determine the association between history of mental illness and ADNM, a Welch's test was performed, since variance homogeneity between groups was not given.

Multiple regression analysis was conducted to determine factors linked to the functional outcome at one year post-operation, with the AOFAS sum score serving as dependent variable. The model was controlled for AOFAS sum score at 4 weeks postoperation. Age, the number of smoked cigarettes per day and fracture type, ADNM sum score, PHQ-9 score and QRI Social Support scale were entered as independent variables. To control for multicollinearity in the model, variance inflation factors were determined for the independent variables and had to be below 10.

p. 4 of 7 S. Weimert et al.

Variables	Baseline Mean (SD)	Four weeks post-operation Mean (SD)	One year post-operation Mean (SD)	F-value	<i>p</i> -value***
ADNM Sum Score	45.1 (14.1)	46.5 (16.2)	40.8 (14.1)**	7.65 ^a	0.003
ADNM Preoccupation	2.0 (0.67)	1.9 (0.7)	1.7 (0.6)**	7.37 ^a	0.399
ADNM Avoidance	1.4 (0.48)	1.4 (0.6)	1.3 (0.6)**	1.62 ^a	0.195
ADNM Failure to adapt	1.5 (0.6)	1.6 (0.7)	1.4 (0.5)**	3.35 ^a	0.030
ADNM Depression	1.5 (0.6)	1.5 (0.5)	1.4 (0.5)**	0.84 ^a	0.446
ADNM Anxiety	1.5 (0.7)	1.6 (0.8)	1.5 (0.7)**	3.47 ^a	0.024
ADNM Impulse control	1.5 (0.7)	1.6 (0.8)	1.3 (0.6)**	2.26 ^a	0.103
PHQ-9	_	5.6 (4.5)*	3.5 (3.6)**	13.14 ^a	0.000
AOFAS Sum	-	58.3 (13.5)	89.3 (14.8)	-18.04 ^b	0.000
AOFAS Pain	-	32.0 (9.0)	35.0 (8.3)	-2.93 ^b	0.005
AOFAS Function	-	18.6 (18.6)	45.6 (6.5)	-27.42 ^b	0.000
AOFAS Alignment	-	7.7 (2.7)	9.0 (2.0)	-3.36 ^b	0.001

Differing sample sizes due to patients not giving information/missing data: **n* = 65, ***n* = 63. ***Comparing mean scores between different measurement points. ^aF-value for repeated measures analysis of variance (ANOVA), using Greenhouse-Geisser Correction. ^bT-value for *t*-test for paired samples. ADNM: Adjustment Disorder New Module; PHQ-9: Patient Health Questionnaire-9; AOFAS: American Orthopedic Foot and Ankle Society; SD: standard deviation.

אסואה. אטוטגווופור סוגטועפו אפש שטטעופ, צווע־ש. דמנפור וופורו ענפגנטווומוופ־ש, אסראס. אוופורמו סורווטףפער דטטרמוע אווגיפ סטכופגץ, סס. גנמועמיע עפיומנטוו.

To determine which symptoms of AD contribute to functional outcome, a linear regression analysis was conducted with the ADNM subscales serving as independent variables. A *p*-value < 0.05 was considered significant. Data were analysed using IBM SPSS Statistics 23.

RESULTS

Demographic variables and baseline characteristics

Patients were in mean 41.9 years old (standard deviation (SD) = 15.2; range: 18–73 years) and 53% (n = 35) were male. Demographic information and baseline characteristics are summarized in Table I. The majority (60.6%) of the cohort sustained a type-44B fracture (Table I). Three patients (4.5%) had an open fracture. The majority of the cohort (97.0%) had an ISS of 4. The mean hospital stay was 8.7 days (Table I). Fifty-one percent of the patients were smokers. Compared with a representative German sample (32), the current sample had higher scores on the social support scale.

Psychological diagnoses

According to the SCID, 13 (12.1%) patients had a history of mental illness. Three patients (4.5%) met the diagnostic criteria of AD 4 weeks and one patient (2%) one year post-operation. Three patients (5%) had a recurrent depressive disorder. None of the patients met the criteria for post-traumatic stress disorder (PTSD).

Symptoms of AD and depression

The ADNM and PHQ-9 scores can be seen in Table II. The severity of the symptoms of an AD peak 4 weeks post-operation. Preoccupation is the most common symptom at all measurement times. Preoccupations are recurring memories and thoughts of the fracture (and the situation in which it was sustained), which occur unintentionally. There was no significant association between a history of mental illness and the ADNM sum score 4 weeks post-operation (t(14.55) = -1.05; p = 0.31).

Functional outcome

The mean AOFAS sum score and subscales all increased between over time (Table II). One year post-operation 47 patients (71.2%) had an excellent functional outcome. Three patients (4.5%) had a poor outcome.

Multivariable analysis of research questions

Linear regression analysis was conducted in order to predict long-term functional outcome. The results are shown in Table III. The extent of symptoms of an AD (β =-0.43; p<0.05) and cigarettes smoked per day (β =-0.28; p<0.05), both at 4 weeks post-operation, were significant predictors in the final model (F (7, 41)=3.59, p<0.01, adjusted R² = 0.27) for functional outcome one year post-operation. Variance inflation factors were all below 3. We conducted another regression analysis without the social support variable to include the 16 patients who were not currently in a relationship. Symptoms of AD and the number of cigarettes smoked per day remained significant predictors. The results can be found in the supplemental material (Table SI¹)

¹https://doi.org/10.2340/16501977-2690

Table III. Regression model for predicting long-term functional outcome (n = 48)

	В	SE	β	Т	<i>p</i> -value
AOFAS sum ^a	0.26	0.14	0.26	-0.55	0.09
Age	-0.06	0.11	-0.07	1.76	0.58
Subgroup AO classification	-1.24	3.29	-0.05	-0.38	0.71
ADNM sum ^a	-0.38	0.18	-0.43	-2.10	0.04
PHQ ^a	5.18	6.70	0.16	0.77	0.44
Number of cigarettes/day	-0.29	0.13	-0.28	0.03	0.04
QRI Social Support	0.11	4.33	0.00	-2.14	0.98

^aAssessed at 4 weeks post-operation.

PHQ-9: Patient Health Questionnaire-9; AOFAS: American Orthopedic Foot and Ankle Society; SD: standard deviation: QRI: Quality of Relationships Inventory.

. ,					
Parameter	В	SE	β	Т	<i>p</i> -value
Preoccupations	-8.25	3.44	-0.38	-2.40	0.02
Avoidance	-0.042	5.07	-0.00	-0.01	0.99
Failure to adapt	-2.38	4.94	-0.12	-0.48	0.63
Depression	-10.78	5.67	-0.38	-1.90	0.06
Anxiety	2.74	3.56	0.15	0.77	0.45
Impulse control	3.06	3.09	0.16	0.99	0.33

Linear regression analysis was conducted in order to predict physical function one year post-operation serving as dependent and the subscales of the ADNM serving as independent variables. The results are shown in Table IV A significant effect was observed for the preoccupations scale only ($\beta = -0.38$; p < 0.05) in the regression model (F (6, 59) = 5.18; adjusted R²=0.28; p<0.01).

DISCUSSION

The prevalence of AD was higher in patients who have sustained a fracture of the ankle or foot than in the general population, in which it is estimated to be 0.9%(33). There appears to be a higher risk of developing AD after a fracture. Compared with a representative German sample, the study cohort reported higher levels of depressive symptoms at all measurement points (34).

The current study identified 2 significant factors that were linked to reduced functional outcome one year post-operation: (i) smoking; and (ii) symptoms of AD. These findings support those of previous studies, showing that smoking is predictive of lower physical functioning after ankle fractures (5) and is associated with a variety of orthopaedic conditions and complications (35, 36). Impairing osteoblast function and hindering collagen synthesis are just 2 examples of the many processes of bone and wound healing on which smoking has an adverse effect (37).

Symptoms of AD were related to reduced long-term functional outcome, while the fracture type was not. In the current study the fracture type was associated only with short-term functional outcome. It is possible that a complex fracture delays functional recovery, but has no effect on the long-term outcome. The findings of the current study underline that psychological factors play a vital role in physical recovery after orthopaedic trauma (38, 39).

The results show that symptoms of AD have a substantial correlation with recovery one year after a fracture, even when they do not meet the diagnostic criteria for a psychological disorder. This does not appear to be related to categorical diagnoses, as shown by the results from the clinical interview, but rather to specific symptoms. The current results support the results of Wegener et al. (39), indicating that "even moderately elevated levels of affective distress may lead to poorer functional outcomes". Remarkably, symptoms of AD correlated with functional outcome, while depressive symptoms did not. Preoccupation was the most common symptom of AD in the study cohort 4 weeks post-operation, and the sole psychological factor that had a significant effect on functional outcome. Preoccupations, which are a main characteristic of AD, are triggered by and refer to the fracture. Hence, a fracture-specific symptom, and not general affective or cognitive symptoms, was significantly related to functional recovery. The symptoms of AD and depression overlap, but AD has a trigger by definition, whereas depression has no clear cause and the symptoms are unspecific. These results highlight that mental preoccupation with the injury hinders good rehabilitation, and anre in agreement with previous studies, which have shown that cognitive occupation and coping (such as catastrophic thinking) play a significant role in recovery after orthopaedic trauma (9, 10).

Surprisingly, social support in the relationship had no effect on functional outcome. Previous research shows that a high number of social relationships, not exclusively romantic, is a protective factor in various health conditions (40), which is a possible explanation for this result.

This study should be interpreted in the light of its limitations. Although dropouts did not differ statistically from the patients who completed the study, there is a potential for systematic differences based on unmeasured characteristics. Caution should be exercised when generalizing results of this study to other trauma populations. In addition, other clinical factors of rehabilitation, such as readmissions and complications, were not taken into account. Due to the study design no causal conclusions can be drawn.

The current study supports approaching patients with ankle and lower leg fractures from a biopsychosocial perspective, and challenges the traditional biomedical model. Surgeons should be aware that psychological and behavioural factors impact treatment success even more than injury characteristics (41). Informing patients about the impact of smoking in the recovery process could lead to better functional outcomes. Given the findings of this study, which indicate that symptoms of AD are a risk for a poor functional recovery, it is vital to carefully assess patients who have sustained ankle or lower leg fractures and, if necessary, initiate psychological interventions. Surgeons would be well advised to screen for psychological distress early in the treatment process, especially as even subclinical levels of symptoms of AD are related to impaired functional recovery, as the current results indicate.

p. 6 of 7 S. Weimert et al.

Journal of Rehabilitation Medicine JRN

To ensure optimal treatment, it is crucial to focus not only on the physical symptoms, but also on possible psychological symptoms.

Surgical techniques have become more successful, and it is now important to also improve other aspects of treatment, taking into account the complex processes of recovery.

ACKNOWLEDGEMENTS

This study is supported by the Graute-Oppermann Foundation. The authors would like to express special thanks to the project coordinators Julia Bechtluft-Sachs, Patricia Lutz, Caroline Neumann and Philipp Winter.

Funding sources. The authors have no conflicts of interest to declare.

REFERENCES

- 1. Court-Brown CM, Caesar B. Epidemiology of adult fractures: a review. Injury 2006; 37: 691–697.
- Kaye JA, Jick H. Epidemiology of lower limb fractures in general practice in the United Kingdom. Inj Prev 2004; 10: 368–374.
- Belcher GL, Radomisli TE, Abate JA, Stabile LA, Trafton PG. Functional outcome analysis of operatively treated malleolar fractures. J Orthop Trauma 1997; 11: 106–109.
- Nilsson GM, Jonsson K, Ekdahl CS, Eneroth M. Unsatisfactory outcome following surgical intervention of ankle fractures. Foot Ankle Surg 2005; 11: 11–16.
- Bhandari M, Sprague S, Hanson B, Busse JW, Dawe DE, Moro JK, et al. Health-related quality of life following operative treatment of unstable ankle fractures - a prospective observational study. J Orthop Trauma 2004; 18: 338–345.
- Castillo RC, Bosse MJ, MacKenzie EJ, Patterson BM, Group LS. Impact of smoking on fracture healing and risk of complications in limb-threatening open tibia fractures. J Orthop Trauma 2005; 19: 151–157.
- Lash N, Horne G, Fielden J, Devane P. Ankle fractures: functional and lifestyle outcomes at 2 years. ANZ J Surg 2002; 72: 724–730.
- Holbrook TL, Anderson JP, Sieber WJ, Browner D, Hoyt DB. Outcome after major trauma: 12-month and 18-month follow-up results from the Trauma Recovery Project. J Trauma 1999; 46: 765–771.
- 9. Nota SP, Bot AG, Ring D, Kloen P. Disability and depression after orthopaedic trauma. Injury 2015; 46: 207–212.
- Vranceanu AM, Bachoura A, Weening A, Vrahas M, Smith RM, Ring D. Psychological factors predict disability and pain intensity after skeletal trauma. J Bone Joint Surg Am 2014; 96: e20.
- Putzke JD, Elliott TR, Richards JS. Marital status and adjustment 1 year post-spinal-cord-injury. J Clin Psychol Med Settings 2001; 8: 101–107.
- Gao Q, Leung A, Reinhardt JD, Zhang F, Liang Q, Chen B, et al. Health-related quality of life of fracture victims four years after the 2008 Sichuan earthquake. J Rehabil Med 2015; 47: 867–872.
- Perry MA, Hudson S, Ardis K. "If I didn't have anybody, what would I have done?": experiences of older adults and their discharge home after lower limb orthopaedic surgery. J Rehabil Med 2011; 43: 916–922.
- Stufkens SA, van den Bekerom MP, Kerkhoffs GM, Hintermann B, van Dijk CN. Long-term outcome after 1822 operatively treated ankle fractures: a systematic review of the literature. Injury 2011; 42: 119–127.

- Crichlow RJ, Andres PL, Morrison SM, Haley SM, Vrahas MS. Depression in orthopaedic trauma patients: prevalence and severity. J Bone Joint Surg Am 2006; 88: 1927–1933.
- Xiao M, Li Ja, Zhang X, Zhao Ze. Factors affecting functional outcome of Sichuan-earthquake survivors with tibial shaft fractures: a follow-up study. J Rehabil Med 2011; 43: 515–520.
- 17. Bachem R, Casey P. Adjustment disorder: a diagnosis whose time has come. J Affect Disord 2018; 227: 243–253.
- Reed GM, Mendonca Correia J, Esparza P, Saxena S, Maj M. The WPA-WHO global survey of psychiatrists' attitudes towards mental disorders classification. World Psychiatry 2011; 10: 118–131.
- Härter M, Woll S, Reuter K, Wunsch A, Bengel J. Recognition of psychiatric disorders in musculoskeletal and cardiovascular rehabilitation patients. Arch Phys Med Rehabil 2004; 85: 1192–1197.
- Weimert S, Kuhn S, Rommens PM, Beutel ME, Reiner I. Psychische Faktoren bei Sprunggelenkfrakturen. Unfallchirurg 2019; 122: 464–468.
- Mitchell AJ, Chan M, Bhatti H, Halton M, Grassi L, Johansen C, et al. Prevalence of depression, anxiety, and adjustment disorder in oncological, haematological, and palliative-care settings: a meta-analysis of 94 interview-based studies. Lancet Oncol 2011; 12: 160–174.
- Müller ME, Allgöwer M, Schneider R, Willenegger H. Manual of internal fixation – techniques recommended by the AO-ASIF group. Berlin: Springer; 1991.
- Kitaoka HB, Alexander IJ, Adelaar RS, Nunley JA, Myerson MA, Sanders M. Clinical rating systems for the anklehindfoot, midfoot, hallux, and lesser toes. Foot Ankle Int 1994; 15: 349–353.
- Ibrahim T, Beiri A, Azzabi M, Best AJ, Taylor GJ, Menon DK. Reliability and validity of the subjective component of the American Orthopaedic Foot and Ankle Society clinical rating scales. J Foot Ankle Surg 2007; 46: 65–74.
- Madeley NJ, Wing KJ, Topliss C, Penner MJ, Glazebrook MA, Younger AS. Responsiveness and validity of the SF-36, Ankle Osteoarthritis Scale, AOFAS Ankle Hindfoot Score, and Foot Function Index in end stage ankle arthritis. Foot Ankle Int 2012; 33: 57–63.
- Maercker A, Einsle F, Köllner V. Adjustment disorders as stress response syndromes: a new diagnostic concept and its exploration in a medical sample. Psychopathology 2007; 40: 135–146.
- Glaesmer H, Romppel M, Brahler E, Hinz A, Maercker A. Adjustment disorder as proposed for ICD-11: dimensionality and symptom differentiation. Psychiatry Res 2015; 229: 940–948.
- Einsle F, Kollner V, Dannemann S, Maercker A. Development and validation of a self-report for the assessment of adjustment disorders. Psychol Health Med 2010; 15: 584–595.
- 29. Löwe B, Spitzer RL, Zipfel S, Herzog W. Gesundheitsfragebogen für Patienten (PHQ-D). Manual und Testunterlagen. Karlsruhe: Pfizer; 2002.
- Kroenke K, Spitzer RL, Williams JB. The PHQ-9: validity of a brief depression severity measure. J Gen Intern Med 2001; 16: 606–613.
- Wittchen H-U, Wunderlich U, Gruschwitz S, Zaudig M. Strukturiertes Klinisches Interview f
 ür DSM-IV (SKID), Achse 1. 1997.
- Reiner I, Beutel M, Skaletz C, Brahler E, Stobel-Richter Y. Validating the German version of the Quality of Relationship Inventory: confirming the three-factor structure and report of psychometric properties. PLoS One 2012; 7: e37380.
- Maercker A, Forstmeier S, Pielmaier L, Spangenberg L, Brahler E, Glaesmer H. Adjustment disorders: prevalence in a representative nationwide survey in Germany. Soc Psychiatry Psychiatr Epidemiol 2012; 47: 1745–1752.

Journal of Rehabilitation Medicine

Factors affecting functional outcome after ankle fractures p. 7 of 7

- Kocalevent R-D, Hinz A, Brähler E. Standardization of the depression screener patient health questionnaire (PHQ-9) in the general population. Gen Hosp Psychiatry 2013; 35: 551–555.
- Harvey E, Agel J, Selznick H, Chapman J, Henley M. Deleterious effect of smoking on healing of open tibia-shaft fractures. Am J Orthoped 2002; 31: 518–521.
- Moghaddam A, Zimmermann G, Hammer K, Bruckner T, Grutzner PA, von Recum J. Cigarette smoking influences the clinical and occupational outcome of patients with tibial shaft fractures. Injury 2011; 42: 1435–1442.
- 37. Porter SE, Hanley EN. The musculoskeletal effects of smoking. J Am Acad Orthop Surg 2001; 9: 9–17.
- Becher S, Smith M, Ziran B. Orthopaedic trauma patients and depression: a prospective cohort. J Orthop Trauma 2014; 28: e242–e246.
- Wegener ST, Castillo RC, Haythornthwaite J, MacKenzie EJ, Bosse MJ, Group ftLS. Psychological distress mediates the effect of pain on function. Pain 2011; 152: 1349–1357.
- Umberson D, Karas Montez J. Social relationships and health: a flashpoint for health policy. J Health Soc Behav 2010; 51: S54–S66.
- MacKenzie EJ, Morris Jr JA, Jurkovich GJ, Yasui Y, Cushing BM, Burgess AR, et al. Return to work following injury: the role of economic, social, and job-related factors. Am J Public Health 1998; 88: 1630–1637.

JRN